Author's response to reviews

Title: Dealing with sickness certification - a survey of problems and strategies among general practitioners and orthopaedic surgeons

Authors:

Britt Arrelov (britt.arrelov@sll.se)
Kristina Alexanderson (kristina.alexanderson@cns.ki.se)
Jan Hagberg (jan.hagberg@cns.ki.se)
Anna Lofgren (anna.lofgren@cns.ki.se)
Gunnar Nilsson (gunnar.nilsson@sll.se)
Sari Ponzer (sari.ponzer@sodersjukhuset.se)

Version: 5 Date: 1 October 2007

Author's response to reviews: see over
Dear Editor,

Attached is a new version of the manuscript “Dealing with sickness certification – a survey of problems and strategies among general practitioners and orthopaedic surgeons”, formatted according to your request.

Below we give a point by point response to the changes requested:

Each affiliation is placed on a separate line.

The word count is deleted from the title page.

Key words are deleted from the abstract page. Abstract is updated in the submission system.

We have changed list of abbreviations to 'Abbreviations'

We have changed 'Competing interest' to 'Competing interests'

Reference 8 now include all the authors.

Reference 29 is now published and we have completed the reference.

Table 2-5 - the full stop from the table titles are removed.

Table 1 is now mentioned before the other tables in the method section.

On behalf of the authors
Britt Arrelöv

Correspondence to:
Britt Arrelöv
CVU, Forum
Box. 175 33
118 91 Stockholm
Sweden